Author’s response to reviews

Title: Delay in the diagnosis and treatment of pulmonary tuberculosis in Uzbekistan: a cross-sectional study

Authors:

Tatiana Belkina (belkinat@faf.cuni.cz)
Doniyor Khojieev (info.mz.rk@minzdrav.uz)
Mirzagaleb Tillyashaykhov (t.mirza58@mail.ru)
Zinaida Tigay (info.mz.rk@minzdrav.uz)
Marat Kudenov (info.mz.rk@minzdrav.uz)
Jurjen Duintjer Tebbens (jurjend@faf.cuni.com)
Jiri Vlcek (vlcek@faf.cuni.cz)

Version: 3  Date: 22 October 2014

Author’s response to reviews: see over
Date: October 22, 2014

Philippa Harris  
Executive Editor  
BMC Infectious Diseases  

Re: Manuscript reference No. 1182631278133247

Dear Dr Harris,

Please find attached a revised version of our manuscript “Delay in the diagnosis and treatment of pulmonary tuberculosis in Uzbekistan: a cross-sectional study”, which we would like to resubmit for publication as a Research article in BMC Infectious Diseases.

In accordance with your suggestion, the revised manuscript was checked by English consultant (Edanz editing). We hope that the revisions in the manuscript will be sufficient to make our manuscript suitable for publication in BMC Infectious Diseases.

Yours faithfully,

Tatiana Belkina  
Department of Social and Clinical Pharmacy  
Faculty of Pharmacy in Hradec Kralove  
Charles University in Prague  
1203, Heyrovsky street  
Hradec Kralove 50005  
Czech Republic